Calculus I, Fall 2004 (http://www.math.nthu.edu.tw/~ wangwc/)

Homework Assignment 10.

Given Dec 09, due Dec 21.

1. Section 6.11: Problems: 12, 14, 26, 28, 30. 2. Section 7.1: Problems: 38. 3. Section 7.2: Problems: 4, 6, 42. 4. Section 7.3: Problems: 45, 46. 5. Section 7.4: Problems: 28, 46, 56, 58.

6. Section 7.8: Read page 499-500, examples 1-3. Then do problems: 3, 4, 10. 7. Section 8.1: Problems: 8, 26, 30, 33, 40, 46, 68(a,b,c), 69(a,d). Remark: You are strongly advised to do as many problems as you can in problems 1-60. Practice pays.